Does a foreign subsidiary's network status affect its innovation activity? Evidence from post-socialist economies

Jože P. Damijan Črt Kostevc Matija Rojec WP06/10

Resumen

Una minuciosa encuesta realizada a 809 filiales en cinco economías post-socialistas (Alemania Oriental, Polonia, Rumania, Eslovenia y Croacia) ha servido de base para estudiar los factores determinantes de la actividad innovadora de las filiales extranjeras. Los datos de la encuesta incluyen los determinantes tradicionales de la actividad innovadora de las empresas así como los indicadores del estatus de la filial extranjera. Nuestros resultados demuestran que las filiales extranjeras son relativamente independientes en cuanto a su actividad innovadora, mientras que al mismo tiempo las filiales con un mejor acceso a los resultados de I+D de las empresas matrices son más propensas a innovar. Se encuentran, sin embargo, importantes diferencias en los factores que determinan los productos y los procesos de innovación: i) las filiales que más invierten en I + D presentan mayor probabilidad para la innovación en productos, pero no en los procesos; ii) la adquisición de conocimientos externos y el tamaño de la empresa tienen un impacto significativo y positivo únicamente en el proceso de innovación; (iii) la transferencia de responsabilidades desde la sede a las filiales y el hecho de que el inversor extranjero sea una empresa multinacional contribuye a la innovación de procesos; (iv) la motivación de búsqueda de mercados por parte de los inversores extranjeros tiene un impacto negativo sobre el nivel de innovación de productos; (v) una mayor antigüedad de la filial es positiva para su proceso de innovación puesto que un inversor extranjero necesita algún tiempo para iniciar las actividades de innovación en una filial.

Palabras clave: Empresas transnacionales, inversión directa, redes, actividad innovadora, economias postsocialistas

Abstract

A detailed questionnaire survey among 809 foreign subsidiaries in five post-socialist economies (East Germany, Poland, Romania, Slovenia and Croatia) is used to study determinants of innovation activity of foreign subsidiaries. Survey data comprise traditional firm innovation activity determinants and indicators of a foreign subsidiary status. Our findings demonstrate that foreign subsidiaries are relatively independent as far as innovation activity is concerned, while at the same time subsidiaries with better access to foreign parent companies R&D results are more likely to innovate. Important differences, however, are found in factors that determine product and process innovation; (i) subsidiaries that invest more in R&D exhibit higher probability for product but not for process innovation; (ii) acquisition of external knowledge and company size have significant and positive impact on on process innovation only, (iii) transfer of responsibilities from headquarters to subsidiaries and foreign investor being a MNE is conducive to process innovation; (iv) marketseeking motivation of foreign investors has a negative impact on product innovation status; (v) higher age of subsidiary is positive for its process innovation, i.e. a foreign investor needs some time to initiate innovation activities in a subsidiary.

Key words: Transnational firms, direct inversion, networks, innovation activity, post-socialist economies.

Paper prepared within the Sixth Framework Programme project "Understanding the Relationship between Knowledge and Competitiveness in the Enlarging European Union (U-Know)".

Jože P. Damijan University of Ljubljana (Faculty of Economics); LICOS, KU Leuven, Belgium.

Črt Kostevc University of Ljubljana (Faculty of Economics).

Matija Rojec

University of Ljubljana (Faculty of Social Sciences); Institute for Macroecopnomic Analysis and Development, Ljubljana.

Instituto Complutense de Estudios Internacionales, Universidad Complutense de Madrid. Campus de Somosaguas, Finca Mas Ferre. 28223, Pozuelo de Alarcón, Madrid, Spain.

© Jože P. Damijan, Črt Kostevc, Matija Rojec

ISBN: 978-84-693-2350-2 Depósito legal:

El ICEI no comparte necesariamente las opiniones expresadas en este trabajo, que son de exclusiva responsabilidad de sus autores/as.

Index

1.	Introduction
2.	Theoretical background and existing empirical evidence on the determinants of foreign subsidiaries' innovation activity
	2.1Determinants of firm innovation activity
3.	Model
4.	Data, sample characteristics and descriptive statistics14
5.	Results17
6.	Conclusions
	Appendix21
	Bibliographical references

1. Introduction

Internationalisation of R&D is one of the prevailing features of the globalisation processes in the last decade. R&D performed in foreign subsidiaries is the most obvious type of the internationalisation of R&D. In 1995-2003, R&D expenditures of foreign subsidiaries in the OECD area increased from USD 34 billion to USD 71 billion, i.e. an increase by 110%, compared to 77% increase of their turnover. In all OECD countries apart from Spain, foreign subsidiaries increased their R&D expenditures by up to four times more than domesticallycontrolled firms. As a result, the share of foreign subsidiaries in total business sector R&D expenditures of OECD has become quite significant. In 2004 this share was in the range between 50%-70% in Ireland, Hungary and Belgium, between 30%-50% in Czech Republic, Sweden, Australia, United Kingdom and Canada, and between 20%-30% in Netherlands, Germany, Italy, France, Portugal, Slovakia (OECD, 2008).

Still, looking at the innovation activity of foreign subsidiaries, their superiority over domestic companies is not obvious. Empirical evidence goes both ways. Falk and Falk (2006) observe that innovation intensity of foreign subsidiaries in Austria is lower than that of domestic firms. Griffith, Redding and Simpson (2004) find that British-owned multinational enterprises (MNEs) account for larger share of R&D activity in UK as compared to foreignowned MNEs, Ebersberger and Lőőf (2005) claim that R&D intensity of domestically owned MNEs in Sweden is significantly higher than in any other ownership types of firms. The situation in Norway and Finland has been the same, except for Anglo-Saxon foreignowned MNEs. Similarly, Almeida and Fernandes (2006) suggest that majority foreignowned firms in developing countries are significantly less likely to engage in technological innovations than minority foreign owned or domestic-owned firms. On the other hand, Damijan, Jaklič and Rojec (2006) for Slovenia, Balcet and Evangelista (2005) for Italy, Gűnther (2006) for East Germany, and Girma, Gong and Görg (2006) for China find that foreign subsidiaries have higher propensity to innovate than domestic companies. One cannot explain much of the difference by foreign ownership itself but rather by the fact that foreign subsidiaries are overrepresented in

higher technology intensive industries, that they are larger in size (Balcet and Evangelista, 2005), more export intensive, have more modern equipment, etc. Once these factors are controlled for, foreign ownership tends not to have significant impact on the propensity to innovate. Thus, according to Gűnther (2006), foreign subsidiaries play a positive role for innovation in a host country since factors, which positively influence innovation – e.g. size, R&D, export intensity, more recent technology – particularly benefit foreign subsidiaries.

The objective of this paper is to ascertain how much does foreign subsidiary's status affect its innovation activity in five post-socialist economies.¹ We build our study on a detailed questionnaire survey of innovation activity among 809 foreign subsidiaries with a particular focus on indicators reflecting foreign subsidiaries status within the MNE network. The existing literature on foreign subsidiaries' innovation activity concentrates on the comparison of innovation activity of foreign subsidiaries and domestically-controlled firms and on a foreign-parent company network as a source of knowledge which may stimulate innovation activity of a subsidiary. The literature that explicitly deals with the determinants of foreign subsidiaries' innovation activity is, however, almost non-existent. The notable exception is Kokko and Kravtsova (2008) who analyse determinants of innovative capability in MNE subsidiaries in four transition countries (Estonia, Hungary, Poland, Slovenia). Therefore, in looking for the theoretical background of foreign subsidiaries' innovation activity one has to refer to various other streams of literature, i.e. those related to determinants of firm innovation activity, knowledge transfer within MNE, subsidiary's position within its foreign parent company network, and heterogeneity of foreign investors and foreign subsidiaries.

This paper tests five propositions. The first is that innovation activity of a foreign subsidiary basically depends on the same factors as that of any other firm. Therefore, the first set of determinants of foreign subsidiaries innovation activity relates to generally identified de-

¹ Of which four are new EU member states (NMS: East Germany, Poland, Romania, Slovenia), while Croatia is a EU candidate country.

terminants of a firm innovation activity. The second proposition is that innovation activity of a foreign subsidiary is co-determined by the extent of knowledge transfer within its foreign-parent company network. This is the first key feature, which differentiates foreign subsidiaries from domestically-controlled firms. It is expected that this knowledge transfer, in principle, improves foreign subsidiary's innovation capacity as compared to a domestically owned firm. The third proposition is that innovation activity of a foreign subsidiary is dependent on a specific position within its foreign parent company network. This is the second key feature, which differentiates foreign subsidiary from domestically-controlled firm. This position can work both ways as far as foreign subsidiary's innovation activity is concerned. The fourth proposition is that various sources of heterogeneity of foreign investors and foreign subsidiaries affect the innovation activity of foreign subsidiaries. The fifth proposition is that host-country and hostmarket characteristics also influence the innovation activity of foreign subsidiaries.

Our empirical strategy is as follows. Compared to related literature, we take into account a much broader set of determinants. We follow the classical approach by explicitly bringing in the empirical model the 'traditional' determinants of firm innovation activity and then gradually adding foreign-subsidiary specific determinants. Our findings confirm that while foreign subsidiaries in the analysed countries are relatively independent in terms of innovation activity - the subsidiaries with better access to foreign parent companies R&D results are more likely to innovate. We find, however, significant differences in factors that determine product and process innovation of subsidiaries. Most notably, market-seeking motivation of foreign investors has a negative impact on product innovation status findings, while transfer of responsibilities from headquarters to subsidiaries is conducive to process innovation.

The remainder of the paper is structured as follows. Section two provides the theoretical background and existing empirical evidence on foreign subsidiaries innovation activity. Section three presents the model, section four describes the data and sample characteristics and descriptive statistics, while section five provides estimations of the model, i.e. of the importance of various sets of determinants on foreign subsidiaries' innovation activity. The last section concludes.

2. Theoretical background and existing empirical evidence on the determinants of foreign subsidiaries' innovation activity

Innovation activity of foreign subsidiaries is essentially determined by two sets of factors. The first set is traditional firm-innovation activity determinants, which are of relevance to any firm regardless of whether foreign or domestically owned. The second, which is the focus of our interest, are the determinants arising from the fact that a firm is a foreign subsidiary. The latter is in the focus of our interest. Within this context one has to refer to various streams of literature. First, to the literature on the determinants of firm innovation activity which relates to the first set. Second, to the literature on knowledge transfer within MNEs, subsidiary's position within its foreign parent company network, heterogeneity of foreign investors and foreign subsidiaries, which altogether relate to the second set of determinants.

The literature that explicitly deals with the determinants of foreign subsidiaries' innovation activity is almost non-existent. The notable exception is Kokko and Kravtsova (2008) who analyse determinants of innovative capability in MNE subsidiaries in Estonia, Hungary, Poland and Slovenia. According to them, the innovative capability of MNE subsidiaries depends on three sets of determinants: (i) the role of the subsidiary in the MNE's international production network; (ii) some other subsidiary characteristics, like size, age, and industry of origin; and (iii) host country and host industry characteristics, including the development level of the host industry and the competitive pressure exerted by local firms. The main finding of Kokko and Kravtsova (2008) is that innovative capability in product and process technology seems to be determined by a different set of variables than capability related to marketing and management knowledge. The most independent subsidiaries are also those that have the strongest innovative capability in product and process technology. At the same time subsidiaries in high technology industries recorded lower levels of innovative capability. For marketing and management capability, the pattern is almost reversed. The highest levels of innovative capability are recorded in subsidiaries that are closely tied to the parent company, with high foreign ownership shares and substantial exports back to the parent company.

2.1 DETERMINANTS OF FIRM INNOVATION ACTIVITY

<u>Proposition one:</u> Innovation activity of a foreign subsidiary depends on the same factors as that of any other firm. Therefore, the first set of determinants of foreign subsidiaries innovation activity comprises of generally identified determinants of firm innovation activity. The issues/factors which are in the centre of attention here are own R&D, external sourcing of knowledge, absorption capacity plus a number of other firm innovation activity determinants.

Own R&D and other standard explanatory variables. Own R&D is the crucial determinant of firm's innovation capacity and of firm's capacity to absorb external knowledge (Cohen and Levinthal, 1989; Romer, 1990; Grossman and Helpman, 1991; Aghion and Howitt, 1992, 1998 etc.). Other most often tested explanatory variables of firm's innovation activity are firm size, industry characteristics, market concentration, technology characteristics, product diversification and market position (export orientation) (Veugelers and Cassiman, 1999; Cohen and Levinthal, 1989). The relationship between innovativeness and size is positive, but not necessarily linear, and it depends on industry characteristics. Industry characteristics are the determinant of firm's innovation activity in the sense that higher-tech industries exhibit higher innovation activity. Industry variable captures several dimensions, i.e. scope for future demand, opportunities for technological innovations and cumulativeness of knowledge, indicating to which extent current innovation build further on previous R&D (Veugelers and Cassiman, 1999: 65). The model of Veugelers and Cassiman (1999: 70-75) for Belgian manufacturing firms confirms that large and more export oriented firms in high tech industries are more likely to innovate, while results of Mairesse and Mohnen (2006) for French manufacturing firms do not fully support these findings. Their main finding is that R&D is positively correlated with all measures of innovation output, and, all other things equal, more correlated than size to innovation. Innovation is generally more sensitive to R&D in low-tech sectors than in the high-tech sectors.

In our approach, own R&D and other standard explanatory variables of firms' innovation activity will be analysed through: (i) share of R&D expenditures in total sales, (ii) type of product with respect to its technological intensity, as a proxy for industry characteristics, (iii) number of employees as a proxy for size.

External sourcing of knowledge. The second set of factors which co-determine firm's innovation activity is external sourcing of knowledge in its various forms, i.e. licensing agreements, contracting out of R&D, buying of equipment (imports), innovation cooperation, knowledge spillovers from other firms and learning-by-exporting. The issue of external sourcing of knowledge is usually considered within the transfer of technology issue, but rarely within the context of innovation-activity determinants. The examples of the latter are Veugelers (1997), Veugelers and Cassiman (1999), Frenz and Ietto-Gilles (2007), and Damijan, Jakli and Rojec (2006). They all recognise that own R&D is the most important for innovation activity, but external sources are also relevant. One of the main conclusions here is that external sourcing does stimulate firm's own R&D but only if a firm possesses adequate absorptive capacity (Veugelers, 1997). This points to the complementarity between in-house R&D and external knowhow (Allen, 1986; Cohen and Levinthal. 1990); it is difficult to be a good 'buyer' when one is not also a 'maker' (Radnor, 1991). That is why Veugelers and Cassiman (1999) find it is more appropriate to talk about 'make and buy' and not 'make or buy' innovation strategy of firms.

The following possible sources of external knowledge will be analysed: (i) acquisition of external knowledge (via licensing agreements and by contracting out R&D activities), (ii) acquisition of external knowledge by importing, (iii) learning by exporting.

Licensing agreements are one of the basic channels of international technology transfer (Eaton and Kortum, 1996), but nowadays provide a less important source, as the latest and most valuable technologies are not available on license (UNCTAD, 2000). As far as contracting out of R&D activities is concerned, Veugelers (1997) claims that it does significantly stimulate own R&D but only when absorptive capacity is present. Imports of intermediate products and capital equipment may also work as a channel of technology transfer (Feenstra, Markusen and Zeile, 1992). Empirical research suggests that imports are an important source of technology, in that much knowledge and R&D is embodied in goods that are imported, especially capital goods and machinery (Hoekman and Smarzynska Javorcik, 2006; Keller and Yeaple, 2003; Keller, 2004; Eaton and Kortum, 2001; Coe and Helpman, 1995; Coe, Helpman and Hoffmaister, 1997; Xu and Wang, 1999; Keller, 2000; Lumenga-Neso, Olarreaga and Schiff, 2001; Keller, 2002; Kraay, Soloaga and Tybout, 2001).

The concept of learning-by-exporting says that more intensive competition on export markets stimulate exporters to strengthen their growth and performance (Van Biesebroeck, 2003). Via their international contacts, exporters acquire knowledge on new production methods, inputs and so forth (Aw, Chen and Roberts, 1998). The literature is not unanimous about the existence of learning-by-exporting effects; they have been found in the case study literature, whereas authors of econometric studies take a much more sceptical view (Keller, 2004). Where only a small minority of authors find support for the learning-by-exporting hypothesis, the vast majority of studies fail to find any evidence of firms' productivity benefiting from exporting activities (see Greenaway and Kneller, 2007 for a survey).

External sourcing of knowledge tends to stimulate firm's own R&D and innovation if a firm possesses adequate absorptive capacity (Veugelers, 1997). Most of the literature on absorption capacity relates to FDI spillovers. The predominant conclusion is that technology spillovers from MNEs tend to occur more frequently when the technological and social capabilities of the host country and the absorptive capacity of firms in the economy are high enough (Blomström, 1986; Kokko, 1994; Kokko, Tansini and Zejan, 1996; Cameron, 1996; Imbriani and Reganati, 1997; Borensztein, De Gregorio and Lee, 1998; Cameron, Proudman and Redding, 1998; Kinoshita, 2000; Keller and Yeaple, 2003; Damijan, Knell, Majcen and Rojec, 2003a; Glass and Saggi, 1998; Girma, Greenaway and Wakelin, 2001; Girma and Gőrg, 2002; Griffith, Redding and Simpson, 2004; Lim, 2001; Halpern and Murakozy, 2006; Abraham, Konings and

Slootmaekers, 2006).² Far the most frequently quoted determinant of absorption capacity is human capital (Borensztein, De Gregorio and Lee, 1998; Hoppe, 2005; Kneller and Stevens, 2006; Gorodnichenko, Svejnar and Terrell, 2006). *In our exercise, absorption capacity will be proxied by the extent of own R&D.*

2.2 KNOWLEDGE TRANSFER WITHIN MNE

<u>Proposition two:</u> Innovation activity of a foreign subsidiary relies upon the extent of knowledge transfer within its foreign parent company network. This is the first key feature, which differentiates foreign subsidiary from a domestically-controlled firm. It is expected that this knowledge transfer, in principle, improves foreign subsidiary's innovation capacity as compared to domestic firms.

The issue of technology and technology transfer from parent companies to their foreign subsidiary, as well as organization of technology, is at the heart of the theories of international production and MNEs (see, for instance, Dunning, 1993; Bartlett, 1986; Bartlett and Ghoshal, 1989; Perez, 1998). In the transaction cost (internalisation) approach (Coase, 1937; Williamson, 1975, 1981, 1985; Buckley and Casson, 1976), one of the main reasons for the existence of international production is the presence of market failures in technological transactions (Williamson, 1981). Technology also has the central position also in the macroeconomic development approach to international production, i.e in the product life cycle hypothesis (Vernon, 1966) and in the flying geese model (Kojima, 1978; Kojima and Ozawa, 1985).

In looking at the theoretical foundations of knowledge transfer within MNEs it seems appropriate to apply resource based, as well as organizational and institutional perspectives. Such an eclectic approach encompasses the whole complexity of the relationship between foreign parent and its subsidiary. The resource-based theoretical perspective focuses on idiosyncratic resources and capabilities of firms as drivers of their performance. Several studies point that FDI in transition economies often requires a massive resource transfer (managerial, financial, knowledge etc) from the foreign parent to the local subsidiary. A

² In contrast, a handful of authors claim that bigger technological gap offers more room for technological spillovers (Findlay, 1978; Haskel, Pereira and Slaughter, 2001; Castellani and Zanfei, 2003).

major stream of literature researches how foreign investors can facilitate organisational learning (Steensma and Lyles, 2000; Lane, Salk and Lyles, 2001) and how this, in turn, influences firm performance (Lyles and Salk, 1996; Dhanaraj, Lyles, Steensma and Tihanyi, 2004). Organisational and institutional theory is the second way to approach knowledge transfer within MNEs. The underlying assumption of knowledge complementarity the concept (Buckley and Carter, 2004) and the dynamic capabilities approach (Teece, Pisano and Shuen, 1997) is that a subsidiary cannot develop idiosyncratic resources nor dynamic capabilities independent from the MNE headquarter or other parts of the MNE. Gupta and Govindarajan (1994) suggest that within the same MNE, strategic roles can be expected to differ in terms of the extent and directionality of knowledge flows between foreign subsidiary and the rest of the corporation (more on that in section 2.3).

Although it is far from guaranteed that all activity carried out by MNEs in host countries will lead to technology transfer, empirical evidence on technology transfer from MNEs to foreign subsidiaries in terms of higher productivity levels and growth is ample.³ On the other hand, empirical evidence on the impact of FDI on foreign subsidiaries' innovation activity is more scarce. According to Cantwell and Molero (2003: 5-7), there is little evidence of any great difference in the innovation behaviour of foreign-owned compared to domestically-owned firms. The difference between the two groups is more a result of structural differences, such as a larger average size of foreign subsidiaries and their greater presence in high-tech sectors (Molero and Heys, 2002). However, innovation strategies require increasingly more global sourcing and parent MNEs more often tend to integrate their subsidiaries in their innovation strategies. More innovation active foreign subsidiaries will tend to source more know-how from local sources and, at the same time, will become more interesting vehicles for technology diffusion to the local economy. But this may also lead to the situation when valuable knowhow leaves the country, while the subsidiary

³ Empirical studies, using firm-level panel data, include Haddad and Harrison, 1993; Blomström and Wolff, 1994; Blomström and Sjöholm, 1999; Aitken and Harrison, 1999; Blalock, 2001; Girma, Greenaway and Wakelin, 2001; Alverez, Damijan and Knell, 2002; Barry, Görg and Strobl, 2002; Girma and Görg, 2002; Damijan, Knell, Majcen, Rojec, 2003b; Arnold and Smarzynska Javorcik, 2005. R&D remains too dependent on the assimilation of know-how developed elsewhere in the parent company (Sanna-Randaccio and Veugelers, 2003: 17-18).

In our empirical approach, knowledge transfer within MNEs will be proxied by the subsidiaries' own judgement of the importance of headquarters R&D for subsidiary's R&D and innovation.

2.3 SUBSIDIARY'S POSITION WITHIN ITS FO-REIGN PARENT COMPANY NETWORK

<u>Proposition three</u>: Innovation activity of a foreign subsidiary is co-determined by its specific position within the parent company network. This is the second key feature, which differentiates foreign subsidiaries from a domesticallycontrolled firm.

Transfer of technology from foreign parent to its subsidiary and innovation activity of a subsidiary inter alia depend on subsidiary's position in parent company's network. The literature on subsidiary's strategy focuses on the process through which foreign subsidiaries enhance their resources and capabilities, and in so doing increasingly add value to the MNE network as a whole (White and Poynter, 1984; Bartlett and Ghoshal, 1989; Young, Hood and Dunlop, 1988; Birkinshaw and Hood, 1998). One of the most widely used subsidiary typologies within this stream of literature is the one by White and Poynter (1984). They distinguish between the following types of subsidiaries: marketing satellite, miniature replica, rationalised manufacturer, specialised producer and strategic independent. The strategic roles of a subsidiary in White and Poynter's typology change according to its technological capacity (Couto, Conceicao Goncalves and Amaral Fortuna, 2003), where moving to more advanced forms of subsidiaries typically involves their higher autonomy and innovative capability (Kokko and Kravtsova, 2008).

Gupta and Govindarajan (1994) suggest that within the same MNE, strategic roles can be expected to differ in terms of the extent and directionality of knowledge flows between foreign subsidiary and the rest of the corporation. They find that innovation by foreign subsidiaries is more typically the results of autonomous initiative rather than strategic directives issued from the corporate headquarter. For example, if a subsidiary operates as a centre of excellence or has been assigned a world product mandate, it is likely to have the autonomy to develop, manufacture, and market a product-line worldwide. Birkinshaw, Hood and Jonsson, (1998) contrast such 'high contributory role subsidiaries' with the implementer and rationalised subsidiary types, which lack autonomy, authority, and capabilities to generate independent competencies (Young and Tavares, 2004). However, the relation between technological capabilities and autonomy in the foreign subsidiary is not clear-cut (Young and Tavares, 2004). On one hand, subsidiaries with greater R&D capabilities may be less technologically dependent on their headquarters and hence display higher levels of autonomy (Birkinshaw and Morrisson, 1995; Pearce, 1999; Taggart and Hood, 1999). On the other hand, strategic sensitiveness of knowledge-related activities can lead to tighter control by headquarters (Bartlett and Ghoshal, 1989; Martinez and Jarillo, 1991). Moreover, strong headquarter assignments may not only facilitate control but also efficiency of the MNE's internal market (Mudambi, 1999; Egelhoff, Gorman and McCormick, 1998).

In our exercise, the position of subsidiaries in their foreign parent companies networks will be analysed via: (i) integration of subsidiary into foreign parent company network via customer-supplier relations of subsidiary with other parts of its MNE group (share of subsidiary's exports/imports to foreign parent company network in total exports/imports), (ii) transfer of responsibilities from foreign investor to its subsidiary, (iii) the division of control between subsidiaries and their foreign parent companies in various business functions.

2.4. HETEROGENEITY OF FOREIGN INVES-TORS AND FOREIGN SUBSIDIARIES

<u>Proposition four</u>: Various other sources of heterogeneity of foreign investors and foreign subsidiaries impact the innovation activity of foreign subsidiaries.

Heterogeneity of foreign investors and foreign subsidiaries may also influence the innovation activity of foreign subsidiaries. Likely the most important source of this heterogeneity are different positions of subsidiaries within their foreign parent companies networks. Still, there are other sources of heterogeneity. To the best of our knowledge there is no literature that would directly tackle the impact of foreign investors and foreign subsidiaries heterogeneity on the latters' innovation activity. However, the issue is broadly covered in the literature on FDI spillovers, which can be usefully applied for our purpose.

The most important sources of foreign subsidiaries heterogeneity relates to domestic versus export market orientation of a subsidiary (Smarzynska, 2003: 6; UNCTAD, 2001; Altenburg, 2000; Belderbos, Campannelli and Fukao, 2001; Sgard, 2001; Tytell and Yudaeva, 2005; Moran, 2005), acquisition versus greenfield type of FDI (UNCTAD, 2001; Belderbos, Campannelli and Fukao, 2001; UNCTAD, 2000; Toth and Semjen, 1999) and joint venture (local equity participation) versus wholly foreign owned subsidiaries (Smeets and de Vaal, 2006; Smarzynska Javorcik and Spatareanu, 2006; Abraham, Konings and Slootmaekers, 2006; Almeida and Fernandes, 2006; Gorodnichenko, Svejnar and Terrell, 2006). Studies on sources of heterogeneity of foreign investors which influence the intensity of FDI spillovers as a rule concentrates on the home country of foreign investors (Abraham, Konings and Slootmaekers, 2006; Perez, 1998; Graham and Krugman, 1989; Levy and Dunning, 1993). Time dimension is another source of heterogeneity in FDI spillovers literature (Kosova, 2006; Cantwell, 1989).

In our exercise, the following sources of firm heterogeneity will be taken into account: (i) subsidiaries' heterogeneity: motivation of foreign investor, share in equity held by foreign investor, (ii) foreign investors' heterogeneity: type of foreign owner, (iii) time dimension: year of entry of foreign investor.

2.5. HOST COUNTRY CHARACTERISTICS

<u>Proposition five</u>: *Host country characteristics also influence the innovation activity of foreign subsidiaries.*

It is widely accepted that host country characteristics have an impact on the type of FDI and foreign subsidiaries. The most obvious way to proxy host country's absorption capacity is to look at its overall level of development (Blomström, Lipsey and Zejan, 1994; Xu, 2000; Kokko and Kravtsova, 2008). Economists often conceive absorptive capacity as a certain level of human capital and education capacity (Borenzstein, De Gregorio and Lee, 1998; Kokko and Kravtsova, 2008); the higher the level of human capital the better are chances for technology transfer via FDI and innovation activity of subsidiaries. Another possible determinant of host country's absorption capacity is investment and business climate. Liberal investment and business climate is more likely to attract more dynamic FDI with more technology transfer etc. (Moran, 1998; Lim, 2001: 4-9; Kokko and Blomström, 1995; Balasubramayam, Salisu and Sapsford, 1996, 1999; Ernst, 1998). Also, Kinoshita and Lu (2006) show that technology spillovers via FDI in developing countries take place only when the host country has the sufficient level of infrastructure. Kokko and Kravstova (2008) quote Rugman and Douglas (1986), Egelhoff, Gorman and McCormick (1998), and Walsh, Linton, Boylan and Sylla (2002) as those who claim that market structure, infrastructure and education are likely to encourage upgrading of affiliates.

In our exercise, a host country dummy will be used to take account of host country characteristics.

3. Model

Based on the above propositions, we estimate the following empirical model (1)

$$\begin{split} P(Inov_{it} = 1) &= \alpha + \beta_1(Inov.det_{it-1}) + \beta_2(know.transf.exter_{it-1}) + \\ & \beta_3(know.transf.inter_{it-1}) + \beta_4(sub.position_{it-1}) + \\ & \beta_5(heterog)_{it-1} + \beta_6(host.country)_{ti} + \varepsilon_{it} \end{split}$$

where $Inov_{it}$ stands for innovatory activity at time *t*. We employ three different variables to account for innovatory activity, namely dummies for product and process innovation as well as information on the share of new or significantly improved products in firm's total sales. Firms' innovatory activity is explained by six sets of explanatory variables:

> a/ Innovation determinants (*Inov.det*_{it}. *i*) represent classical determinants of innovation, where we use the share of R&D expenditure in total sales and firm size, as measured by total employment. We expect both variables to have a positive impact on subsidiaries' innovatory activity.

> b/ External sources of knowledge are included under *know.transf. exter*_{t-1}. In order to control for possible sources of knowledge outside the firm and the

MNE network, we employ the share of imports in total supplies, the share of exports in total sales (learning-byexporting) as well as subsidiaries' own opinion on the importance of acquisition and purchase of external knowledge for their innovatory activity. We expect all three variables to have a positive impact on subsidiaries' innovatory activity.

c/ Importance of knowledge transfer within the MNE network (know.transf.interit) is estimated with the inclusion of the observed importance of R&D activities performed at the MNE headquarters for subsidiaries' innovatory activity. We expect headquarters R&D to have a positive impact on subsidiaries' innovatory activity.

d/ *sub.position*_{t-1} serves as a measure of the role of subsidiary's position within the MNE network on subsidiary's innovatory activity. Subsidiary's position within the MNE network is, firstly, measured by the share of subsidiary's exports to other parts of the network in its overall exports and the share of imports from network in overall imports. This helps indicate the reliance of a subsidiary on selling/buying from the network and, consequently, on its integration in the network. Secondly, we introduce subsidiaries' perception of the degree of transfer of new responsibilities (related to new geographical markets, products and business functions) from headquarters to a subsidiary. Thirdly, subsidiaries' perception of the degree of control they have over operational, marketing and strategic business functions is included. We expect the higher the subsidiaries' responsibilities and control over business functions the higher their innovatory activity.

e/ Heterogeneity of subsidiaries and foreign investors is captured under heterogit and proxied by: (i) strategic motivation of foreign investors, which is divided into market-seeking, efficiency-seeking and strategic-assetseeking, (ii) foreign share in total equity as a proxy for foreign control, (iii) year of entry of foreign investor to control for length of foreign presence, and (iv) different types of foreign investors, i.e. multinational (MNE) groups (present in several countries and having above 250 employees or EUR 50 million in turnover), small and medium-sized foreign firm, and foreign financial investors. We expect efficiency and strategic-asset seeking FDI to have a positive impact on innovatory capacity and the opposite for market-seeking FDI. We also expect MNEs as foreign parents to have a positive impact on innovatory capacity.

f/ host country is a host country dummy, used to take account of host country characteristics.

All regressors are lagged one period to mitigate the issues of endogeneity. Dependent variables therefore represent values for year 2006, while our independent variables in the preferred regression represent the 2005 values. In the Appendix we present results with a longer lag for the regressor variables as they taken on values from 2002 instead. The results do not differ significantly from results with one period lags, therefore only focus on the former.

4. Data, sample characteristics and descriptive statistics

To evaluate the impact of different groups of determinants of innovation on both innovation status and innovation intensity of firms we use data from the IWH FDI Micro Database. The Database encompasses 809 manufacturing (NACE Rev 1.1, industries 15 to 37) foreign subsidiaries (firms with 10 per cent or higher foreign equity share) with about 214,000 employees from five countries, i.e. Croatia, East Germany, Poland, Romania and Slovenia, which were surveyed by a comprehensive questionnaire in 2007. The selection of countries tries to balance country size, geographic location and level of economic development. The data relate to 2005, for some variables the data for 2002 and 2005 were collected, and for some for 'at entry' and 'today'. The survey questionnaire was centrally designed and followed the same pattern in all five countries. Altogether a population of 6,833 firms were approached, of which 11.84 per cent responded. In terms of employment, the response rate was 19.05%.

In terms of the number of firms the three largest sectors in the total population are food products and beverages, wearing apparel and dressing, and fabricated metal. In terms of employment the three largest sectors are again food and wearing apparel, and this time manufacturing of motor vehicles, trailers & semitrailers. The deviations of the sample from the distribution of the population across sectors are up to 3 per cent, if we consider the number of firms, and up to 5 per cent if we take the employment into account. In terms of distribution of firms across size classes, the sample is underrepresented for micro (1-9) and small enterprise (10-49), and consequently overrepresents medium seized (50 - 249) and large (above 250) firms.

Table 1 below presents the shares of foreign subsidiaries questioned undertaking product and process innovations. About two thirds of the firms surveyed claimed they have made product innovations in the past three years. Similarly, about two thirds of firms claimed to have introduced process innovations. Country wise, the share of product innovators firms is the highest in Poland, followed by East Germany - West Germany (EG-WG) MNEs, East Germany and Slovenia. Croatia and Romania lag somewhat behind. As far as process innovations is concerned, Slovenia is in the lead, followed by EG-WG MNEs, Poland and East Germany. Again, Croatia and Romania lag behind.⁴ There is a considerable difference of sample foreign subsidiaries' innovatory activity according to the type of their foreign parent companies. 60.7% of subsidiaries with MNEs as foreign parent companies claimed that they were innovatory active in the last three years, while the corresponding shares for small or mediums sized firms is 27.1% and for financial investors only 4.3%.

⁴ Foreign subsidiaries from the sample seem to be much more innovatory active than enterprises in the analysed countries on average. According to Eurostat (2008: 103), the share of innovatory enterprises in all enterprises is 24.8% in Poland, 26.9% in Slovenia, 19.5% in Romania.

		Share of product innova-	Share of process innova-
Country	No. of firms	tors	tors
East Germany	222	68.9%	68.9%
EG-WG MNEs	73	76.7%	74.0%
Croatia	144	59.0%	60.4%
Poland	110	79.1%	70.9%
Romania	220	57.7%	57.3%
Slovenia	40	65.0%	75.0%
Total	809	67.1%	66.3%

 Table 1: Share of innovatory firms among surveyed foreign subsidiaries

Source: IWH FDI Micro Database and own calculations.

Note: Product innovators are firms that succeeded in introducing new or significantly improved products or services in the past three years. Product must be new to the surveyed firm, not necessarily to the market. Process innovations are new or improved production or delivery methods including e.g. changes in techniques, equipment and/or software.

The numeric variables, most of which appear as regressors in (1) are described in Table 2. The only variable in Table 2 that serves as a regressant is the approximate share of new and significantly improved products in total firm sales (in %), where the mean and median values are relatively low, but display a very high standard deviation. Still, if one faces high share of product innovators among the surveyed firms (67.1%) with much lower average share of new products in the surveyed firms sales (14.3%, with median value of only 5.0%), it is obvious that new or significantly new products only rather gradually gain importance for subsidiaries overall activity.

In addition to firm size, as measured by the average number of employees in 2005, we also use the share of R&D expenditures in total sales as standard determinants of innovation. On average, surveyed firms spend 4.8% of

their sales for R&D, with the median value of only 0.8%. Importance of acquisition of external knowledge is an ordinal variable, whereby the respondents were asked to rank its importance for the firm from 1-not important to 5extremely important. Both the average and median values for this variable are close to 2.5, which is in the middle between 2-little important and 3-important. On average, the surveyed subsidiearies are highly export and import oriented, with 50.6% share of exports (median value is 50%) and 45.8% share of imports (median value is 40%) in sales. The importance of R&D undertaken by the MNE headquarters was, again, assessed on an ordinal scale ranging from 1-not important to 5important. Here the average values (as measured by the mean and median) are near to 3important, i.e. somewhat higher than was the case with acquiring external knowledge.

Table 2: Descriptive statistics of numeric variables						
Variable	Ν	Mean	Median	Std. Dev.	Min	Max
Share of new products in total firm sales (in %)	633	14.250	5.0	37.251	0	100
Standard	determ	inants of ir	nnovation			
R&D expenditure share in total sales (in %)	662	4.830	0.8	11.23392	0	80
Total employment (No. of employees)	803	266.781	101	532.9778	1	6505
Extern	al sour	ces of know	vledge			
Importance of acquisition of external knowledge (from 1=not important to 5=extremely important)	446	2.482	2.5	1.180227	1	5
Share of imports in total supplies (in %)	772	45.767	40	35.75004	0	100
Share of exports in total sales (in %)	780	50.591	50	38.53494	0	100

Table 2: Descriptive statistics of numeric variables

Knowle	edge tra	ansfer withir	MNE			
Importance of head quarters R&D (from 1=not important to 5=extremely important)	717	2.796	3	1.462888	1	5
Subsidiary's	s positi	on within MI	NE netwo	rk		
Share of exports to network in total exports (in %)	780	7.935	1.7	12.15206	0	100
Share of imports from network in total imports (in %)	660	39.711	28.6	40.15911	0	100
Transfer of responsibilities from head- quarters (from 1=no transfer to 4=full transfer)	722	1.871	1.7	0.884919	1	4
Operational control (from 1=fully con- trolled by subsidiary to 4=fully con- trolled by foreign investor)	786	1.570	1	0.83143	1	4
Marketing control (from 1=fully con- trolled by subsidiary to 4=fully con- trolled by foreign investor)	771	2.314	2	1.11311	1	4
Strategic control (from 1=fully con- trolled by subsidiary to 4=fully con- trolled by foreign investor)	720	2.326	2.4	0.881323	1	4
	Hete	rogeneity				
Market-seeking strategy (from 1=not important to 5=extremely important)	758	2.815	3	1.095932	1	5
Efficiency-seeking strategy (from 1=not important to 5=extremely important)	767	3.020	3	1.313073	1	5
Strategic-asset seeking strategy (from 1=not important to 5=extremely impor- tant)	751	2.521	2.5	1.027249	1	5
Share of foreign equity (in %)	797	87.885	100	22.19562	0	100
Year of foreign investor entry	787	1997.546	1998	4.738325	1970	2006

As far as subsidiary's position within MNE network is concerned, may be the most outstanding feature is a low share of subsidiaries' exports to other affiliates within the MNE network relative to overall exports, which is dwarfed by much higher share of imports from the network in total imports (the respective mean values are approximately 8% and 40%). Obviously, subsidiaries are much more integrated in their foreign parent companies' network on the supplies than on the sales side. In other words, subsidiaries do not seem to produce intermediate products for other parts of their network, but products for arms length buyers. Thus, within their foreign parents' networks, subsidiaries are responsible for certain markets and/or for certain products. These responsibilities seem to be related to subsidiaries right at the entry of foreign investor, as since then only quite limited transfer of responsibilities has happened. The extent to which responsibilities have been transferred to subsidiaries since the entry of foreign investor, was measured on an ordinal scale ranging from 1-no transfer to 4-full transfer. The interviewees assess the extent of transfer as being on average under the range of 'limited transfer'.

Yet another indicator of subsidiaries position within foreign parent companies' networks is division of control over individual operational, marketing and strategic business functions.⁵ This indicator was also measured on an ordinal scale ranging from 1=fully controlled by subsidiary to 4=fully controlled by foreign investor. As expected, foreign investors are more eager to control strategic and marketing business functions than the operational ones.

⁵ Operational control is proxied by the interviewees' perception of the division of control between subsidiary and foreign parent company as far as production and operational management is concerned. Marketing control is proxied by control over market research and marketing, while strategic control is proxied by control of a combination of business functions, i.e. basic and applied research, product development, process engineering, strategic management and planning, investment project and finance.

Within the 'heterogeneity' variables we look into motivation of foreign investors, foreign share in total equity and age. Wholly foreign owned subsidiaries dominate, while the average foreign share in the equity is 87.9%. Foreign investors' motives have been grouped into market-seeking, efficiency seeking and strategic-asset seeking ones. The importance of these is evaluated by the subsidiaries on an ordinal scale from 1-not important to 5extremely important. Efficiency-seeking motivation seems to be the most frequent, closely followed by market-seeking and the least frequent being strategic-asset-seeking motivation. Average value of strategic-asset-seeking motive, which serves as a proxy for being motivated by an acquisition of subsidiary's knowledge, is in the range between 2-little important and 3-important.

5. Results

We estimate (1) by using a simple probit in cases where the dependent variables are either the probability to product innovate (columns 1 and 4) or the probability to process innovate (columns 3 and 6), while we employ standard OLS for the case when the dependent variable is share of new or significantly different products in total sales (columns 2 and 5). Results are presented in Table 3. Columns 1 to 3 present estimates without industry dummies, while columns 4-5 include industry dummies.

Results presented in Table 3 reveal several key findings. Within all the analyzed sets of determinants (standard determinants of innovation, external sources of knowledge, knowledge transfer within MNE, subsidiary's position within MNE network and heterogeneity) there are individual determinants which significantly impact the innovatory activity of foreign subsidiaries:

> al Standard determinants of innovation. One of the key determinants of innovation is expectedly the share of R&D expenditure in sales indicating that firms that spend more on R&D are more likely to product innovate and will be able to base a larger share of their sales on newly developed products. This, however, is not the case for process innovation, where firm size exhibits statistically significant impact (albeit with a very low value of the ratio) on the likelihood of innovation. This is likely related to the fact that larger scale producers tend to benefit more from process innovation than smaller firms. Compared with low tech industries, more advanced industries do not seem to be more likely to innovate, on the contrary, when industry dummies are included, high-tech firms are even significantly less likely to process innovate than low-tech firms.

	Product	Share of	Process	Product	Share of	Process
	innova-	new	innova-	innova-	new	innova-
	tion	products	tion	tion	products	tion
	status	-	status	status	-	status
	1	2	3	4	5	6
Sta	andard dete	erminants o	f innovatio	n		
R&D expenditure share in total	0.031*	0.033***	0.006	0.029*	0.033***	0.004
sales	[0.017]	[0.012]	[0.012]	[0.016]	[0.013]	[0.013]
Total employment	0.0001	0.0001	0.0004**	0.0001	0.0000	0.0005**
	[0.0001]	[0.0001]	[0.0002]	[0.0001]	[0.0001]	[0.0002]
High-tech industry dummy	0.474	0.409	0.217	0.02	0.178	-1.278*
	[0.485]	[0.489]	[0.528]	[0.716]	[0.786]	[0.734]
Medium-tech industry dummy	0.108	0.416**	-0.061	-0.284	0.713	-1.09
	[0.214]	[0.212]	[0.212]	[0.513]	[0.733]	[0.843]
	External so	ources of ki	nowledge			
Importance of acquisition of	0.108	0.177	0.202*	0.151	0.213*	0.174
external knowledge	[0.108]	[0.115]	[0.108]	[0.116]	[0.120]	[0.119]
Share of imports in total supplies	-0.006*	-0.006**	-0.005	-0.005	-0.005	-0.006*
	[0.003]	[0.003]	[0.003]	[0.004]	[0.004]	[0.004]
Share of exports in total sales	-0.001	-0.002	-0.001	-0.001	-0.002	0.001
-	[0.004]	[0.004]	[0.004]	[0.004]	[0.004]	[0.004]

Table 3: Impact of affiliate characteristics on innovation activity

	Knowledge	transfer w	ithin MNE			
Importance of headquarters	0.209**	0.082	0.135*	0.236***	0.095	0.163*
R&D	[0.082]	[0.085]	[0.080]	[0.084]	[0.088]	[0.085]
	idiary's pos					
Share of exports to network in	0.001	-0.003	-0.009	0.002	-0.004	-0.006
total exports	[0.011]	[0.012]	[0.012]	[0.012]	[0.012]	[0.012]
Share of imports from network in	-0.001	0.000	-0.003	-0.002	0.094	-0.003
total imports	[0.003]	[0.003]	[0.003]	[0.003]	[0.297]	[0.003]
Transfer of responsibilities from	0.091	0.006	0.275**	0.108	0.001	0.322***
headquarters	[0.109]	[0.113]	[0.109]	[0.113]	[0.121]	[0.112]
Operational control	0.192	0.146	0.053	0.211	0.178	0.054
	[0.134]	[0.141]	[0.131]	[0.139]	[0.152]	[0.135]
Marketing control	-0.053	0.094	0.206	-0.046	0.095	0.193
	[0.142]	[0.149]	[0.142]	[0.147]	[0.153]	[0.145]
Strategic control	-0.252	-0.263	-0.21	-0.28	-0.272	-0.12
-	[0.189]	[0.206]	[0.184]	[0.195]	[0.214]	[0.192]
	He	eterogeneity	y			
Market-seeking strategy	-0.142	0.127	-0.046	-0.177*	0.094	0.011
	[0.089]	[0.094]	[0.088]	[0.099]	[0.100]	[0.095]
Efficiency-seeking strategy	0.060	0.019	0.121	0.072	0.031	0.091
	[0.077]	[0.079]	[0.081]	[0.081]	[0.087]	[0.079]
Strategic-asset seeking strat.	0.134	0.027	0.030	0.138	0.017	0.018
	[0.098]	[0.098]	[0.099]	[0.104]	[0.108]	[0.104]
Share of foreign equity	-0.003	0.006	-0.001	-0.003	0.007	-0.002
	[0.004]	[0.005]	[0.005]	[0.005]	[0.005]	[0.005]
Owner foreign MNE	-0.144	0.096	0.523**	-0.096	0.070	0.545**
	[0.245]	[0.259]	[0.239]	[0.257]	[0.277]	[0.258]
Owner small or medium sized	-0.154	0.390	0.185	-0.171	0.330	0.261
enterprise	[0.250]	[0.259]	[0.242]	[0.260]	[0.276]	[0.250]
Owner financial investor	0.151	0.607	0.137	0.242	0.711	0.370
	[0.489]	[0.521]	[0.480]	[0.475]	[0.519]	[0.490]
Year of foreign-investor entry	-0.024	-0.048**	-0.035	-0.037*	-0.059**	-0.047**
	[0.020]	[0.021]	[0.022]	[0.022]	[0.023]	[0.023]
Host-country dummies	YES	YES	YES	YES	YES	YES
Industry dummies	NO	NO	NO	YES	YES	YES
Observations	256	239	256	250	237	254
Pseudo R-squared [#]	0.13	0.17	0.17	0.16	0.20	0.23

Note: Dependent variable in columns 1 and 4 is an indicator variable of product innovation, dependent variable in columns 2 and 5 is the share of sales attributed to a new product, while in columns 3 and 6 it is a process innovation indicator variable. Robust standard errors in brackets.

* significant at 10%; ** significant at 5%; *** significant at 1%.

[#] except columns 2 and 5, where we report the adjusted R-squared.

b/ External sources of knowledge. Although the importance of acquiring external knowledge has a positive impact on the probability of innovation and its measurable impact, the coefficient is only significant for process innovation when industry dummies are excluded and for the share of new products with industry dummies. On the contrary, there is some evidence that firms with a higher share of imports in total supplies will be less likely to innovate and will also benefit less from innovation in terms of the share of new products in total sales. c/ *Knowledge transfer within MNE*. R&D activities of the foreign parent company headquarters seem to be quite important for subsidiaries' likelihood to innovate. Headquarters' R&D activities have positive and significant impact on subsiadiries' innovation activity with the ratio of 0.163 for process and 0.236 for product innovations, if industry dummies are included.

d/ *Subsidiary's position within MNE network* does not really seem to have a very important impact on subsidiary's innovation activity. Of all the variables

tested within this set of determinants, it is only the transfer of responsibilities from headquarters to subsidiaries, which is conducive to process innovation. Division of control between headquarters and subsidiaries, or subsidiaries trade with the network do not prove to have any impact on subsidiaries' innovatory activity. All these, as well as low share of subsidiaries' exports going to parent network, seem to indicate that foreign subsidiaries in new EU member states function as fairly independent entities as far as innovatory activity is concerned. Their innovatory activity is motivated by the increase of their own competitiveness and not by some broader parent company goals.

e/ Heterogeneity. Market-seeking motivation of foreign investors has a negative impact on product innovation status. In other words, FDI looking for the local market are not conducive to product innovation activity of the subsidiaries. Still, there is no indication that efficiency or strategic assetseeking FDI would have a positive impact on subsidiaries innovation activity. On the other hand, it is important who is a foreign investor, since it is only MNEs, and not small or medium sized enterprises or financial investors, who have significant and positive impact on process innovation status of subsidiaries. Also, the length of foreign presence in a subsidiary has a positive impact on innovation and also the share of new products in total sales. It therefore seems that a foreign investor needs some time to initiate innovation activities in a subsidiary. Foreign share in equity, as a measure of foreign control, does not impact subsidiary's innovation activity. This is in line with the lack of impact of division of control of business functions on the innovation activity (see above).

6. Conclusions

From a host country point of view, increased R&D and innovation activity of foreign subsidiaries means more opportunities for knowledge transfer in the host economy; firstly directly to the subsidiaries under foreign ownership and control and, secondly, indirectly to other firms in the host economy through spillovers. Therefore, identification of the determinants of foreign subsidiaries' innovation activity is of direct relevance for host country policy makers. In other words, apart from FDI spillovers, attracting of R&D intensive FDI with high innovation capabilities is definitely the most legitimate reason for a government to promote inward FDI. According to the latest CREST report (European Union, 2008: 12), FDI in R&D is high on the political agenda of most EU member states, although the R&D part is usually included in more general FDI polices. Although only a limited number of countries have specific policy instruments in place to stimulate spillovers from FDI in R&D there is a rising awareness to innovate policy measures in order: (i) to take advantage of inward FDI in R&D by means of embedding (former) high-tech enclaves with little knowledge diffusion in the local environment and to generate spillovers without hollowing out the local research base; (ii) to capture the scientific benefits of outward FDI in R&D (back) to domestic R&D environments; (iii) to adapt policy measures to the rationale of knowledge competition rather than cost competition.

Our findings confirm that, compared to average firms in the analysed countries, surveyed foreign subsidiaries are much more innovative in terms of the fraction of product and process innovators in overall number of firms than the average firms. Roughly two thirds of the surveyed subsidiaries undertook product and process innovations in the last three years, Still, new or significantly new products only rather gradually gain importance for subsidiaries overall activity, as their share in total sales is only 14.3% in terms of mean and 5.0% in terms of median value. Taking account of a wide array of information on standard determinants of innovation, external sources of knowledge, knowledge transfer within MNE, subsidiary's position within MNE network and heterogeneity, we come to several interesting findings.

First, subsidiary's position within MNE network does not really seem to have a very important impact on subsidiary's innovation activity. Only, transfer of responsibilities from headquarters to subsidiaries is conducive to process innovation, while division of control between headquarters and subsidiaries, or subsidiaries trade with the network do not prove to have any impact on subsidiaries' innovatory activity. Foreign subsidiaries in new EU member states seem to be relatively independent ventures as far as innovatory activity is concerned.

Second, there are differences in factors that determine product and process innovation. Subsidiaries with higher R&D expenditures and more transfer of R&D results from headquarters do more product innovations; company size and acquisition of external knowledge do not impact product innovation activity. Situation with process innovations is different. While R&D activities of the headquarters remain to have positive and significant impact on subsidiaries' innovatory activity, the size of own R&D expenditures does not. However acquisition of external knowledge and company size have significant and positive impact on process innovation, while subsidiaries in high-tech industries exhibit significantly lower probability to process innovate than low-tech firms.

Third, market-seeking motivation of foreign investors has a negative impact on product innovation status, but there is no indication that efficiency or strategic asset-seeking FDI would have a positive one. The fact that foreign investor is a MNE, and not a small or medium sized enterprise or a financial investor, is positive for subsidiary's process innovation. The same goes for the age of subsidiary, i.e. a foreign investor needs some time to initiate innovation activities in a subsidiary.

activity wit	<u>h longer la :</u>		<u>control v</u>	ariables		
	Product	Share of	Process	Product	Share of	Process
	innovation	new	innovation	innovation	new	innovation
	status	products 2	status 3	status 4	products 5	status
				•	5	6
	ndarad dete					
R&D expenditure share in total	0.027	0.04	0.004	0.021	0.044	0.003
sales	[0.018]	[0.018]**	[0.014]	[0.018]	[0.020]**	[0.015]
Total employment	0.000	0.000	0.001	0.000	0.000	0.001
	[0.000]	[0.000]	[0.000]*	[0.000]	[0.000]	[0.000]**
High-tech industry dummy	0.173	0.1	0.093	-0.244	0.175	-0.861
	[0.495]	[0.483]	[0.499]	[0.751]	[0.781]	[0.833]
Medium-tech industry dummy	-0.149	0.383	-0.272	-0.457	-0.09	-1.983
	[0.223]	[0.232]*	[0.227]	[0.596]	[0.883]	[0.975]**
	External so					
Importance of acquisition of	0.123	0.263	0.193	0.146	0.34	0.166
external knowledge	[0.122]	[0.132]**	[0.128]	[0.131]	[0.151]**	[0.136]
Share of imports in total supplies	-0.64	-0.625	-0.01	-0.449	-0.542	-0.335
	[0.350]*	[0.354]*	[0.357]	[0.397]	[0.424]	[0.411]
Share of exports in total sales	0.068	0.029	-0.402	0.024	-0.081	-0.071
·	[0.401]	[0.405]	[0.413]	[0.470]	[0.456]	[0.481]
	Knowledge					
Importance of headquarters	0.197	0.046	0.100	0.208	0.026	0.12
R&D	[0.091]**	[0.089]	[0.089]	[0.097]**	[0.092]	[0.100]
	idiary's pos					
Share of exports to network in	-0.008	-0.009	-0.002	-0.01	-0.015	0.002
total exports	[0.012]	[0.013]	[0.013]	[0.013]	[0.013]	[0.013]
Share of imports from network in	0.206	0.218	-0.327	0.096	0.355	-0.348
total imports	[0.290]	[0.318]	[0.306]	[0.300]	[0.351]	[0.326]
Transfer of responsibilities from	0.119	-0.114	0.281	0.132	-0.153	0.371
headquarters	[0.120]	[0.120]	[0.122]**	[0.127]	[0.133]	[0.132]***
Operational control	0.049	0.19	-0.182	0.046	0.25	-0.223
	[0.146]	[0.152]	[0.145]	[0.150]	[0.162]	[0.149]
Markating apptral	0.077	0.051	0.117	0.109	0.076	0.08
Marketing control	[0.159]					
<u>Stratagia control</u>		[0.161]	[0.158]	[0.162]	[0.167]	[0.159]
Strategic control	-0.227	-0.264	0.113	-0.25	-0.284	0.284
	[0.216]	[0.221]	[0.213]	[0.224]	[0.230]	[0.226]
Manlast as alive a starte m		terogeneit		0.007	0.004	
Market-seeking strategy	-0.157	0.032	-0.134	-0.207	-0.004	-0.102
	[0.105]	[0.101]	[0.104]	[0.119]*	[0.107]	[0.110]
Efficiency-seeking strategy	0.049	0.012	0.164	0.081	0.008	0.121
	[0.086]	[0.084]	[0.094]*	[0.093]	[0.095]	[0.092]
Strategic-asset seeking strat.	0.158	0.03	0.076	0.169	0.009	0.083
	[0.103]	[0.100]	[0.107]	[0.110]	[0.115]	[0.114]
Share of foreign equity	-0.003	0.004	-0.009	-0.003	0.005	-0.01
	[0.005]	[0.005]	[0.005]	[0.005]	[0.006]	[0.006]*
Owner foreign MNE	-0.05	0.373	0.487	0.029	0.398	0.462
	[0.271]	[0.257]	[0.265]*	[0.297]	[0.290]	[0.294]
Owner small or medium sized	-0.242	0.469	0.005	-0.199	0.498	0.075
enterprise	[0.279]	[0.268]*	[0.273]	[0.295]	[0.297]*	[0.293]
Owner financial investor	-0.546	0.332	-0.264	-0.405	0.336	-0.331
	[0.620]	[0.627]	[0.605]	[0.652]	[0.563]	[0.640]
Year of foreign-investor entry	-0.035	-0.014	-0.033	-0.053	-0.026	-0.044
J,	[0.025]	[0.026]	[0.027]	[0.027]*	[0.027]	[0.030]
Host-country dummies	YES	YES	YES	YES	YES	YES
Industry dummies	NO	NO	NO	YES	YES	YES
				113	113	163

Appendix: Impact of affiliate characteristics on innovation activity with longer lags on the control variables

Observations	214	209	214	208	207	212
Pseudo R-squared [#]	0.13	0.14	0.19	0.15	0.21	0.26

 Note: Dependent variable in columns 1 and 4 is an indicator variable of product innovation, dependent variable in columns 2 and 5 is the share of sales attributed to a new product, while in columns 3 and 6 it is a process innovation indicator variable.

 Robust standard errors in brackets.

 * significant at 10%; ** significant at 5%; *** significant at 1%.

 # except columns 2 and 5, where we report the adjusted R-squared.

Bibliographical references

Abraham, F., J. Konings and V. Slootmaekers (2006), *FDI Spillovers, Firm Heterogeneity and Degree of Ownership from Chinese manufacturing.* Catholic University Leuven, Mimeo.

Aghion, P. and P. Howitt (1992), "A Model of Growth through Creative Destruction", *Econometrica*, 60: 323-351.

Aghion, P. and P. Howitt (1998), *Endogenous Growth Theory*. Cambridge, MA: MIT Press.

Aitken, B.J. and A.E. Harrison (1999), "Do Domestic Firms Benefit from Direct Foreign Investments? Evidence from Venezuela", *American Economic Review*, 89: 605-618.

Allen, T (1986), Managing the Flow of Technology. Cambridge, MA: MIT Press.

Almeida, R. and A.M. Fernandes (2006), "Openness and Technological Innovations in Developing Countries: Evidence from Firm-Level Surveys". *World Bank Policy Research Working Paper No. WPS3985*.

Altenburg, T. (2000), "Linkages and Spillovers between Transnational Corporations and Small and Medium-Sized Enterprises in Developing Countries: Opportunities and Best Policies", In UNCTAD, TNC-SME Linkages for Development: Issue-Experiences-Best Practices. Geneva: United Nations.

Alverez, I., J.P. Damijan, and M. Knell (2002), *Do Spanish Firms Get Technology through FDI and Trade?* University of Madrid, Madrid. Mimeo.

Arnold, J.M. and B. Smarzynska Javorcik (2005), "Gifted Kids or Pushy Parents? Foreign Acquisitions and Plant Productivity in Indonesia", *World Bank Policy Research Working Paper, No. 3597.*

Aw, B.Y., X. Chen and M.J. Roberts (1998), "Productivity and the Decision to Export: Micro Evidence from Taiwan and South Korea", *NBER Working Paper No. 6558.*

Balasubramanyan, V.N., M. Salisu and D. Sapsford (1996), "Foreign Direct Investment and Growth in EP and IP Countries". *Economic Journal*, 106: 92-105.

Balasubramanyan, V.N., M. Salisu and D. Sapsford (1999), "Foreign Direct Investment as an Engine of Growth", *Journal of International Trade and Economic Development*, 8: 27-40.

Balcet, G. and R. Evangelista. (2005), "Global Technology: Innovation Strategies of Foreign Affiliates in Italy", *Transnational Corporations*, 14: 53-92.

Barry, F., H. Görg, and E. Strobl (2002), *Productivity Spillovers and Labour-Market Crowding Out: Interactions between Foreign and Domestic Firms in Irish Manufacturing*, University College Dublin. Mimeo.

Bartlett, C.A. (1986), "Managing and Building the Transnational: The New Organizational Challenge. In M.E", Porter ed., *Competition in Global Industries*, Cambridge: Harvard Business School Press. 367-401.

Bartlett, C.A. and S. Ghoshal (1989), *Managing across Borders: The Transnational Solution*. Cambridge, Harvard Business School Press.

Belderbos, R., G. Campannelli and K. Fukao (2001), "Backward Vertical Linkages of Foreign Manufacturing Affiliates: Evidence from Japanese Multinationals", *World Development*, 29(1): 189-208.

Birkinshaw, J and N. Hood (eds.) (1998), *Multinational Corporate Evolution and Subsidiary Development*, New York: St Martins Press.

Birkinshaw, J., N. Hood, and S. Jonsson (1998), "Building Firm-Specific Advantages in Multinational Corporations: the Role of Subsidiary Initiative". *Strategic Management Journal*, 23(4): 773-795.

Birkinshaw, J.M. and A. Morrison (1995), "Configurations of Strategy and Structure in Subsidiaries of Multinational Corporations", *Journal of International Business Studies*, 26: 729-754.

Blalock, G. (2001), *Technology from Foreign Direct Investment: Strategic Transfer through Supply Chains*, University of California, Berkeley. Mimeo.

Blomström, M. (1986), "Foreign Investment and Productive Efficiency: the Case of Mexico". *Journal of In- dustrial Economics*, 15: 97-110.

Blomström, M., R.E. Lipsey and M. Zejan (1994), "What Explains Developing Country Growth?" *NBER Working Paper No. 4132*. National Bureau of Economic Research, Cambridge, MA.

Blomström, M. and F. Sjöholm (1999), "Technology Transfer and Spillovers: Does Local Participation with Multinationals Matter?" *European Economic Review*, 43: 915-923.

Blomström, M. and E. Wolff (1994), "Foreign investment enterprises and productivity convergence in Mexico", In W. Baumol, R. Nelson, and E. Wolff, eds., *Convergence of Productivity: Cross-national Studies and Historical Evidence*, Oxford: Oxford University Press.

Borensztein, E., J. De Gregorio, and J.W. Lee (1998), "How Does Foreign Direct Investment Affect Economic Growth?" *Journal of International Economics*, 45: 115-135.

Buckley, P. J. and M. Carter (2004), "A Formal Analysis of Knowledge Combination in Multinational Enterprises", Journal of International Business Studies, 35: 371-384.

Buckley, P.J. and M. Casson (1976), "The Future of the Multinational Enterprise", London: Macmillan.

Cameron, G. (1996), *Catchup and Leapfrog between the USA and Japan. Chapter 6 of D. Phil. Thesis*, Nuffield College, University of Oxford, Oxford. Mimeo.

Cameron, G., J. Proudman, and S. Redding (1998), "Productivity Convergence and International Openness". In J. Proudman and S. Redding (eds.), *Openness and Growth*, London: Bank of England. Chapter 6.

Cantwell, J. (1989), "Technological Innovation and Multinational Corporations. Oxford: Basil Blackwell.

Cantwell, J. and J. Molero (2003), Introduction. In J. Cantwell and J. Molero (eds.), "*Multinational Enterprises, Innovative Strategies and Systems of Innovation*". Cheltenham: Edward Elgar. p.p.: 1-14.

Castellani, D. and A. Zanfei (2003), "Technology Gaps, Absorptive Capacity and the Impact of Inward Investments on Productivity of European Firms", *Economics of Innovation and New Technology*, 12: 555-576.

Coase, R. H. (1937), "The Nature of the Firm", *Economica*, 4: 386-405.

Coe, D. and E. Helpman (1995), "International R&D Spillovers", *European Economic Review*, 39: 859-887.

Coe, D., E. Helpman, and A. Hoffmaister (1997), "North-South R&D Spillovers", *Economic Journal*, 107: 134-149.

Cohen W. M. and D.A. Levinthal (1989), "Innovation and Learning: The Two Faces of R&D", *Economic Journal*, 99: 569-596.

Cohen, W.M. and D.A. Levinthal (1990), "Absorptive capacity: A new perspective on learning and innovation", *Administrative Science*, 35, 128-52.

Couto, J.P., V.F. da Conceicao Goncalves and M.J. Amaral Fortuna (2003), "Strategic Choice of the Subsidiaries: Contextual and Operational factors", *Journal of Comparative International Management*, 6 (1): 57-70.

Damijan, J., A. Jaklič and M. Rojec (2006), "Do External Knowledge Spillovers Induce Firms' Innovations? Evidence from Slovenia". In A.T. Tavares and A. Teixeira (eds.), *Multinationals, Clusters and Innovation: Does Public Policy Matter?* Basingstoke: Palgrave, 27-47.

Damijan, J.P., M. Knell, B. Majcen and M. Rojec (2003a), "The Role of FDI, R&D Accumulation and Trade in Transferring Technology to Transition Countries: Evidence from Firm Panel Data for Eight Transition Countries", *Economic Systems*, 27: 189-204.

Damijan, J.P., M. Knell, B. Majcen and M. Rojec (2003b), "Technology Transfer through FDI in Top-10 Transition Countries: How Important are Direct Effects", *Horizontal and Vertical Spillovers*". *William Davidson Working Paper No. 549 (February)*. Ann Arbor: The University of Michigan Business School.

De Loecker, J. (2005) "Do Exports Generate Higher Productivity? Evidence from Slovenia", *Working Paper presented at EIIE Conference*. Ljubljana.

Dhanaraj, C., M.A. Lyles, H.K. Steensma and L. Tihanyi (2004) "Managing Tacit and Explicit Knowledge Transfer in IJVs : The Role of Relational Embeddedness and the Impact on Performance", *Journal of International Business Studies*, 35: 428-442.

Dunning, J.H. (1993)" Multinational Enterprises and the Global Economy", Wokingham: Addison Wesley.

Eaton, J. and S. Kortum (1996), "Trade in Ideas: Patenting and Productivity in the OECD", *Journal of International Economics*, 40: 251-271.

Eaton, J. and S. Kortum (2001), "Trade in Capital Goods", *European Economic Review*, 45: 1195-1235.

Ebersberger, B. and H. Lőőf (2005), "Corporate Innovation Activities, Does Ownership Matter? *The CIS Study Statistical Report*", Nordic Innovation Centre's Forum for Innovation Policies, Oslo.

Egelhoff, W., L. Gorman and S. McCormick (1998), "Using Technology as a Path to Subsidiary Development. In J. Birkinshaw and N. Hood (eds.), *Multinational Corporate Evolution and Subsidiary Development*", New York: St Martins Press, 213–238.

Ernst, D. (1998), Globalization, Convergence, and Diversity: "The Asian Production Networks of Japanese Electronic Firms. In D. Ernst and S. Haggard (eds.), *Rivalry or Riches: International Production Networks in Asia*", Ithaca, NY: Cornell University Press.

European Union (2008), "*CREST Report. Internationalisation of R&D. Facing the Challenges of Globalisation: Approaches to a Proactive International Policy in S&T*", Brussels: Scientific and Technical Research Committee.

Falk, M. and R. Falk (2006), "Do Foreign-Owned Firms Have a Lower Innovation Intensity than Domestic Firms?" *WIFO Working Papers No. 275*, Austrian Institute of Economic Research, Vienna.

Feenstra, R., J. Markusen, and W. Zeile (1992), "Accounting for Growth with New Inputs: Theory and Evidence", *American Economic Review*, 82: 415-421.

Findlay, R. (1978), "Relative Backwardness, Direct Foreign Investment, and the Transfer of Technology: A Simple Dynamic Model", *Quarterly Journal of Economics*, 92: 1-15.

Frenz, M. and G. Ietto-Gilles (2007), *"The relevance of internal and external sources of knowledge in the innovation performance of UK enterprises*, EIBA Conference.

Girma, S., Y. Gong and H. Gőrg (2006), *"Can you Teach Old Dragons New Tricks? FDI and Innovation Activity in Chinese State-Owned Enterprises*, University of Nottingham, Nottingham. Mimeo.

Girma, S. and H. Görg (2002), "Foreign Direct Investment, Spillovers and Absorption Capacity: Evidence from Quantile Regressions", *GEP ResearcH Paper 02/14*, University of Nottingham, Nottingham.

Girma, S., D. Greenaway, and K. Wakelin (2001), "Who Benefits from Foreign Direct Investment in the UK?", *Scottish Journal of Political Economy*, 48: 119-133.

Glass, A. and K. Saggi (1998), "International Technology Transfer and the Technology Gap", *Journal of Development Economics*, 55: 369-398.

Gorodnichenko, Y., J. Svejnar and K. Terrell (2006), *Vertical and Horizontal FDI spillovers in Transition Economies: Do Institutions Matter?*, Mimeo.

Graham, E.M. and P.R. Krugman (1989), "*Foreign Direct Investment in the United States*", Washington, D.C.: Institute for International Economics.

Greenaway, D. and R. Kneller (2007), "Firm heterogeneity, exporting and foreign direct investment". *The Economic Journal*, 117: 134-161.

Griffith, R., S. Redding and H. Simpson (2004), "Foreign Ownership and Productivity: New Evidence from the Service Sector and the R&D Lab", *CEP Discussion Paper No 649 (September)*, Centre for Economic Performance, London School of Economics and Political Science, London.

Grossman, G. and E. Helpman (1991), "Innovation and Growth in the Global Economy", Cambridge: MIT Press.

Günther, J. (2006), "Determinants of Innovation – Does Foreign Ownership Matter in Eastern Germany". Paper presented at *EACES* 9^{th} *Bi-Annual Conference: Development Strategies – A comparative View*, 7-9 September, Brighton.

Gupta, A.K. and V. Govindarajan (1994), "Organizing for Knowledge Flows within MNCs", *International Business Review*, 3(4): 443-457.

Haddad, M. and A. Harrison (1993), "Are There Positive Spillovers from Direct Foreign Investments? Evidence from Panel Data for Morocco", *Journal of Development Economics*, 42: 51-74.

Halpern, L. and B. Murakozy (2006), *Does Distance Matter in Spillover?*, Hungarian Academy of Sciences, Institute of Economics, CEU Department of Economics, Budapest. Mimeo.

Haskel, J., S. Pereira and M. Slaughter (2001), "Does Inward Foreign Direct Investment Boost the Productivity of Domestic Firms?", Paper presented at the *NBER Summer Institute*. August.

Hoekman, B. and B. Smarzynska Javorcik (2006), "Lessons from Empirical Research on International Technology Diffusion through Trade and Foreign Direct Investment", In B. Hoekman and B. Smarzynska Javorcik eds., *Global Integration and Technology Transfer*. Washington, D.C.: World Bank, 1-26.

Hoppe, M. (2005), "Technology Transfer through Trade", Nota di Lavoro 19.2005. Fondazione Eni Enrico Mattei, Milano.

Imbriani, C. and F. Reganati (1997), "International Efficiency Spillovers into the Italian Manufacturing Sector – English Summary", *Economia Internazionale*, 50: 583-595.

Keller, W. (2002), "Trade and the Transmission of Technology", Journal of Economic Growth, 7: 5-24.

Keller, W. (2004), *International Technology Diffusion*, University of Texas, National Bureau of Economic Research, Centre for Economic policy Research. Mimeo.

Keller, W. And S.R. Yeaple (2003), "Multinational Enterprises, International Trade, and Productivity Growth: Firm Level Evidence from The United States", *NBER Working Paper No. 9504 (February)*. National Bureau of Economic Research, Cambridge, MA.

Kinoshita, Y. (2000), *R&D and Technology Spillovers via FDI: Innovation and Absorptive Capacity*, CERGE-EI, Prague. Mimeo.

Kinoshita, Y. and C-H. Lu (2006), "On the Role of Absorptive Capacity: FDI Matters to Growth", *William Davidson Institute Working Paper No. 845 (August)*. Ann Arbor: University of Michigan.

Kneller, R. and P.A. Stevens, (2006), "Frontier Technology and Absorptive Capacity: Evidence from OECD Manufacturing Industries", *Oxford Bulletin of Economics and Statistics*, 68 (1): 1-21.

Kojima. K. (1978), "*Direct Foreign Investment : a Japanese Model of Multinational Business Operations*", London: Croom Helm.

Kojima, K. and T. Ozawa (1985), "Toward a Theory of Industrial Restructuring and Dynamic Comparative Advantages", *Hitotsubashi Journal of Economics*, 26: 1-20.

Kokko, A. (1994), "Technology, Market Characteristics, and Spillovers", *Journal of Development Economics*, 43: 279-293.

Kokko, A. and M. Blomström (1995), "Policies to Encourage Inflows of Technology through Foreign Multinationals", *World Development*, 23: 459-468. Kokko, A. and V. Kravtsova (2008), "Innovative capability in MNC subsidiaries: Evidence from four European transition economies", *Post-Communist Economies*, 20 (1): 57-75.

Kokko, A., R. Tansini and M. Zejan (1996), "Local Technological Capability and Spillovers from FDI in the Uruguayan Manufacturing Sector", *Journal of Development Studies*, 34: 602-611.

Kosova, R. (2006), *Do Foreign Firms Crowd Out Domestic Firms: Evidence from the Czech Republic,* (March) Available at SSRN: http://ssrn.com/abstract=891776

Kraay, A., I. Soloaga and J. Tybout (2001), *Product, Quality, Productive Effciency, and International Technology Diffusion: Evidence from Plant-Level Panel Data,* Paper presented at the NBER Summer Institute.

Lane, P.J., E.J. Salk and M.A. Lyles (2001), "Absorptive Capacity, Earning, and Performance in International Joint Ventures", *Strategic Management Journal*, 22: 1139-1161.

Levy, D. and J.H. Dunning (1993), "International Production and Sourcing: Trends and Issues", *STI Review*, No. 13: 13-59.

Lim, E-G. (2001), "Determinants of, and the Relation Between Foreign Direct Investment and Growth: A Summary of the Recent Literature", *IMF Working Paper WP/01/175*.

Lumenga-Neso, O., M. Olarreaga and M. Schiff (2001), *On 'Indirect' Trade-related R&D Spillovers,* World Bank, Washington, D.C., Mimeo.

Lyles, M.A. and J.E. Salk (1996), "Knowledge Acquisition from Foreign Parents in IJVs: An Empirical Examination in the Hungarian Context", *Journal of International Business Studies*, 29: 154-174.

Mairesse, J. and P. Mohnen (2006), "The Importance of R&D for Innovation: A Reassessment using French Survey Data", *NBER Working paper* No. 10897.

Martinez, J. and J.C. Jarillo (1991), "Coordination Demands of International Strategies", *Journal of International Business Studies*, 22: 429-444.

Molero J. and J. Heys (2002), "Differences in Innovative Behaviour between National and Foreign Firms: Measuring the Impact of Foreign Firms on National Innovation Systems", *International Journal of Entrepreneurship and Innovation Management*, 2: 2-3.

Moran, T. (1998), "Foreign Direct Investment and Development", Washington, D.C.: Institute for International Economics.

Moran, T. (2005), "How Does Foreign Direct Investment Affect Host Country Development: Do We Already Know the Answer? Using Industry Case Studies to Make Reliable Generalizations", In T. Moran, E. Graham and M. Blomström eds., *Does Foreign Direct Investment Promote Development?* Washington, D.C.: Institute for International Economics. 281-314.

Mudambi, R. (1999), "MNE Internal Capital Markets and Subsidiary Strategic Independence", *International Business Review*, 8: 197–211.

OECD (2008), "*Recent Trends in the Internationalisation of R&D in the Enterprise Sector*", DSTI/EAS/IND/SWP(2006)1/FINAL. Paris: OECD, Directorate for Science, Technology and Industry.

Pearce, R. (1999), "The Evolution of Technology in Multinational Enterprises: The Role of Creative Subsidiaries", *International Business Review*, 8: 125-148.

Perez, T. (1998), "*Multinational Enterprises and Technological Spillovers*", Amsterdam: Harwood Academic Publishers.

Radnor, M. (1991), "Technology acquisition strategies and processes: a reconsideration of the make versus buy decision", *International Journal of Technology Management*, 6, 113-135.

Romer, P.M. (1990), "Endogenous Technological Change", Journal of Political Economy, 98, S71-S102.

Rugman, A.M. and S. Douglas (1986), "The Strategic Management of Multinationals and World Product Mandating", *Canadian Public Policy*, 12 (2): 320-328.

Sanna-Randaccio, F. and R. Veugelers (2003), "Global Innovation Strategies of MNEs: Implications for Host Economies", In J. Cantwell and J. Molero (eds.), *Multinational Enterprises, Innovative Strategies and Systems of Innovation.* Cheltenham: Edward Elgar. p.p.: 14-46.

Sgard, J. (2001), "Direct Foreign Investments and Productivity Growth in Hungarian Firms", 1992-1999. *Document de travail No. 01.19.* Paris: CEPII - Centre d'edutes prospectives et d'informations internationales.

Smarzynska, B.K. (2003), "Does Foreign Direct Investment Increase the Productivity of Domestic Firms?" In *Search of Spillovers through Backward Linkages. William Davidson Working Paper No. 548.* Ann Arbor: University of Michigan Business School.

Smarzynska Javorcik, B. and M. Spatareanu (2006), *To Share or Not To Share: Does Local Participation Matter for Spillovers from Foreign Direct Investment?* Mimeo.

Smeets, R. and A. de Vaal (2006), *Knowledge Spillovers and FDI Ownership*. Nijmegen School of Management, Radboud University Nijmegen, Nijmegen. Mimeo.

Steensma, H.K. and M.A, Lyles (2000), "Explaining IJV Survival in a Transition Economy through Social Exchange and Knowledge-Based Perspectives", *Strategic Management Journal*, 21: 831-851.

Taggart, J. and N. Hood (1999), "Determinants of Autonomy in Multinational Corporation Subsidiaries", *European Management Journal*, 22: 237-250.

Teece, D.J., G. Pisano and A. Shuen (1997), "Dynamic Capabilities and Strategic Management", *Strategic Management Journal*, 18: 509-533.

Toth, I. and A. Semjen (1999), "Market Links and Growth Capacity of Enterprises in a Transforming Economy: The Case of Hungary", In I.J. Toth and A. Semjen eds., *Market Links, Tax Environment and Financial Discipline of Hungarian Enterprises.* Budapest: Institute of Economics, Hungarian Academy of Sciences.

Tytell, I. and K. Yudaeva (2005), "The Role of FDI in Eastern Europe and New Independent States: New Channels for the Spillover Effect", *Working Paper No. 60*. Center for Economic and Financial research (CE-FIR), Moscow.

UNCTAD (2000), World Investment Report 2000, Geneva: United Nations.

UNCTAD (2001), World Investment Report 2001, Geneva: United Nations.

Van Biesebroeck, J. (2003), "Exporting Raises Productivity in Sub-Saharan African Manufacturing Plants", *NBER Working Paper No. 10020*.

Vernon, R. (1966), "International Investment and International Trade in the Product Cycle", *Quarterly Journal of Economics*, 80: 190-207.

Veugelers, Reinhilde. (1997), "Internal R&D expenditures and external technology sourcing", *Research Policy*, 26 (1997): 303-315.

Veugelers, Reinhilde and Bruno Cassiman (1999), "Make and buy in innovation strategies: evidence from Belgian manufacturing firms", *Research Policy*, 28: 63-80.

Walsh, S., J. Linton, R. Boylan, and C. Sylla (2002), "The Evolution of Technology Management Practice in Developing Economies: Findings from Northern China", *International Journal of Technology Management*, 24 (2/3): 311-329.

White, R. E. and T. A. Poynter (1984), "Strategies for Foreign-Owned Subsidiaries in Canada", *Business Quarterly*, 49(2): 59-69.

Williamson, O. E. (1975), "Markets and Hierarchies: Analysis and Antitrust Implications" New York: Free Press.

Williamson, O.E. (1981), "The Modern Corporation: Origins, Evolution, Attributes", *Journal of Economic Literature*, 19: 1537-1568.

Williamson, O.E. (1985), "The Economic Institutions of Capitalism", New York: Free Press.

Xu, B. (2000), "Multinational Enterprises, Technology Diffusion, and Host Country Productivity Growth", *Journal of Development Economics*, 62: 477-493.

Xu, B. and J. Wang (1999), "Capital Goods Trade and R&D Spillovers in the OECD", *Canadian Journal of Economics*, 32: 1258-1274.

Young, S., N. Hood and S. Dunlop (1988), "Global strategies, multinational subsidiary roles and economic impact in Scotland", *Regional Studies*, 22 (6): 487-497.

Young, S. and A.T. Tavares (2004), "Centralization and Autonomy: Back to the Future", *International Business Review*, 13: 215-237.

Últimos títulos publicados

DOCUMENTOS DE TRABAJO "EL VALOR ECONÓMICO DEL ESPAÑOL"

- **DT 14/10** Antonio Alonso, José; Gutiérrez, Rodolfo: *Lengua y emigración: España y el español en las migraciones internacionales.*
- **DT 13/08** de Diego Álvarez, Dorotea; Rodrigues-Silveira, Rodrigo; Carrera Troyano Miguel: *Estrategias para el Desarrollo del Cluster de Enseñanza de Español en Salamanca*
- **DT 12/08** Quirós Romero, Cipriano: *Lengua e internacionalización: El papel de la lengua en la internacionalización de las operadoras de telecomunicaciones.*
- **DT 11/08** Girón, Francisco Javier; Cañada, Agustín: *La contribución de la lengua española al PIB y al empleo: una aproximación macroeconómica.*
- DT 10/08 Jiménez, Juan Carlos; Narbona, Aranzazu: *El español en el comercio internacional.*
- **DT 09/07** Carrera, Miguel; Ogonowski, Michał: *El valor económico del español: España ante el espejo de Polonia.*
- DT 08/07 Rojo, Guillermo: *El español en la red.*
- **DT 07/07** Carrera, Miguel; Bonete, Rafael; Muñoz de Bustillo, Rafael: *El programa ERASMUS en el marco del valor económico de la Enseñanza del Español como Lengua Extranjera.*
- **DT 06/07** Criado, María Jesús: *Inmigración y población latina en los Estados Unidos: un perfil sociodemográfico.*
- **DT 05/07** Gutiérrez, Rodolfo: *Lengua, migraciones y mercado de trabajo.*
- **DT 04/07** Quirós Romero, Cipriano; Crespo Galán, Jorge: *Sociedad de la Información y presencia del español en Internet.*
- DT 03/06 Moreno Fernández, Francisco; Otero Roth, Jaime: *Demografía de la lengua española.*
- DT 02/06 Alonso, José Antonio: *Naturaleza económica de la lengua.*
- DT 01/06 Jiménez, Juan Carlos: *La Economía de la lengua: una visión de conjunto.*

WORKING PAPERS

- **WP 06/10** P. Jože; Kostevc, Damijan, Črt; Rojec, Matija: *Does a foreign subsidiary's network status affect its innovation activity? Evidence from post-socialist economies.*
- **WP 05/10** Garcimartín, Carlos; Rivas Luis; García Martínez, Pilar: On the role of relative prices and capital flows in balance-of-payments constrained growth: the experiences of Portugal and Spain in the euro area.
- **WP 04/10** Álvarez, Ignacio; Luengo, Fernando*: Financiarización, empleo y salario en la UE: el impacto de las nuevas estrategias empresariales.*
- **WP 03/10** Sass, Magdolna: *Foreign direct investments and relocations in business services what are the locational factors? The case of Hungary.*
- WP 02/10 Santos-Arteaga, Francisco J.: *Bank Runs Without Sunspots*.
- WP 01/10 Donoso, Vicente; Martín, Víctor: *La sostenibilidad del déficit exterior de España.*

WP 14/09	Dobado, Rafael; García, Héctor: <i>Neither so low nor so short! Wages and heights in eight-</i> <i>eenth and early nineteenth centuries colonial Hispanic America.</i>
WP 13/09	Alonso, José Antonio: Colonisation, formal and informal institutions, and development.
WP 12/09	Álvarez, Francisco: <i>Oportunity cost of CO2 emission reductions: developing vs. developed</i> economies.
WP 11/09	J. André, Francisco: <i>Los Biocombustibles. El Estado de la cuestión.</i>
WP 10/09	Luengo, Fernando: <i>Las deslocalizaciones internacionales. Una visión desde la economía crítica</i>
WP 09/09	Dobado, Rafael; Guerrero, David: <i>The Integration of Western Hemisphere Grain Markets in the Eighteenth Century: Early Progress and Decline of Globalization.</i>
WP 08/09	Álvarez, Isabel; Marín, Raquel; Maldonado, Georgina: <i>Internal and external factors of com-</i> <i>petitiveness in the middle-income countries.</i>
WP 07/09	Minondo, Asier: Especialización productiva y crecimiento en los países de renta media.
WP 06/09	Martín, Víctor; Donoso, Vicente: <i>Selección de mercados prioritarios para los Países de Renta Media.</i>
WP 05/09	Donoso, Vicente; Martín, Víctor: <i>Exportaciones y crecimiento económico: estudios empíri-</i> cos.
WP 04/09	Minondo, Asier; Requena, Francisco: ¿Qué explica las diferencias en el crecimiento de las exportaciones entre los países de renta media?
WP 03/09	Alonso, José Antonio; Garcimartín, Carlos: <i>The Determinants of Institutional Quality. More on the Debate.</i>
WP 02/09	Granda, Inés; Fonfría, Antonio: <i>Technology and economic inequality effects on interna-</i> <i>tional trade.</i>
WP 01/09	Molero, José; Portela, Javier y Álvarez Isabel: <i>Innovative MNEs' Subsidiaries in different domestic environments.</i>
WP 08/08	Boege, Volker; Brown, Anne; Clements, Kevin y Nolan Anna: ¿Qué es lo "fallido"? ¿Los Estados del Sur,o la investigación y las políticas de Occidente? Un estudio sobre órdenes políticos híbridos y los Estados emergentes.
WP 07/08	Medialdea García, Bibiana; Álvarez Peralta, Nacho: <i>Liberalización financiera internacional, inversores institucionales y gobierno corporativo de la empresa</i>
WP 06/08	Álvarez, Isabel; Marín, Raquel: <i>FDI and world heterogeneities: The role of absorptive ca-</i> pacities
WP 05/08	Molero, José; García, Antonio: Factors affecting innovation revisited
WP 04/08	Tezanos Vázquez, Sergio: The Spanish pattern of aid giving
WP 03/08	Fernández, Esther; Pérez, Rafaela; Ruiz, Jesús: <i>Double Dividend in an Endogenous Growth Model with Pollution and Abatement</i>
WP 02/08	Álvarez, Francisco; Camiña, Ester: Moral hazard and tradeable pollution emission permits.
WP 01/08	Cerdá Tena, Emilio; Quiroga Gómez, Sonia: Cost-loss decision models with risk aversion.
WP 05/07	Palazuelos, Enrique; García, <i>Clara: La transición energética en China.</i>
WP 04/07	Palazuelos, Enrique: <i>Dinámica macroeconómica de Estados Unidos: ¿Transición entre dos recesiones?</i>

WP 03/07	Angulo, Gloria: <i>Opinión pública, participación ciudadana y política de cooperación en España.</i>
WP 02/07	Luengo, Fernando; Álvarez, Ignacio: Integración comercial y dinámica económica: España
WP 01/07	ante el reto de la ampliación. Álvarez, Isabel; Magaña, Gerardo: ICT and Cross-Country Comparisons: A proposal of a new composite index.
WP 05/06	Schünemann, Julia: <i>Cooperación interregional e interregionalismo: una aproximación so-cial-constructivista.</i>
WP 04/06	Kruijt, Dirk: América Latina. Democracia, pobreza y violencia: Viejos y nuevos actores.
WP 03/06	Donoso, Vicente; Martín, Víctor: <i>Exportaciones y crecimiento en España (1980-2004): Cointegración y simulación de Montecarlo.</i>
WP 02/06	García Sánchez, Antonio; Molero, José: <i>Innovación en servicios en la UE: Una aproximación a la densidad de innovación y la importancia económica de los innovadores a partir de los datos agregados de la CIS3.</i>
WP 01/06	Briscoe, Ivan: Debt crises, political change and the state in the developing world.
WP 06/05	Palazuelos, Enrique: <i>Fases del crecimiento económico de los países de la Unión Europea-15.</i>
WP 05/05	Leyra, Begoña: Trabajo infantil femenino: Las niñas en las calles de la Ciudad de México.
WP 04/05	Álvarez, Isabel; Fonfría, Antonio; Marín Raquel: <i>The role of networking in the competitive-</i> <i>ness profile of Spanish firms.</i>
WP 03/05	Kausch, Kristina; Barreñada, Isaías: <i>Alliance of Civilizations. International Security and Cosmopolitan Democracy.</i>
WP 02/05	Sastre, Luis: An alternative model for the trade balance of countries with open economies: the Spanish case.
WP 01/05	Díaz de la Guardia, Carlos; Molero, José; Valadez, Patricia: <i>International competitiveness in services in some European countries: Basic facts and a preliminary attempt of interpretation.</i>
WP 03/04	Angulo, Gloria: <i>La opinión pública española y la ayuda al desarrollo</i> .
WP 02/04	Freres, Christian; Mold, Andrew: <i>European Union trade policy and the poor. Towards improving the poverty impact of the GSP in Latin America.</i>
WP 01/04	Álvarez, Isabel; Molero, José: <i>Technology and the generation of international knowledge spillovers. An application to Spanish manufacturing firms.</i>
POLICY PAPE	RS

- **PP 01/10** Molero, José: *Factores críticos de la innovación tecnológica en la economía española.*
- **PP 03/09** Ferguson, Lucy: *Analysing the Gender Dimensions of Tourism as a Development Strategy.*
- PP 02/09 Carrasco Gallego ,José Antonio: *La Ronda de Doha y los países de renta media.*
- **PP 01/09** Rodríguez Blanco, Eugenia: *Género, Cultura y Desarrollo: Límites y oportunidades para el cambio cultural pro-igualdad de género en Mozambique.*
- **PP 04/08** Tezanos, Sergio: *Políticas públicas de apoyo a la investigación para el desarrollo. Los casos de Canadá, Holanda y Reino Unido*

- **PP 03/08**Mattioli, Natalia Including Disability into Development Cooperation. Analysis of Initiatives
by National and International Donors
- PP 02/08 Elizondo, Luis: Espacio para Respirar: El humanitarismo en Afganistán (2001-2008).
- **PP 01/08** Caramés Boada, Albert: *Desarme como vínculo entre seguridad y desarrollo. La reintegración comunitaria en los programas de Desarme, desmovilización y reintegración (DDR) de combatientes en Haití.*
- **PP 03/07** Guimón, José: *Government strategies to attract R&D-intensive FDI*.
- PP 02/07 Czaplińska, Agata: Building public support for development cooperation.
- **PP 01/07** Martínez, Ignacio: *La cooperación de las ONGD españolas en Perú: hacia una acción más estratégica.*
- **PP 02/06** Ruiz Sandoval, Erika: *Latinoamericanos con destino a Europa: Migración, remesas y codesarrollo como temas emergentes en la relación UE-AL.*
- **PP 01/06** Freres, Christian; Sanahuja, José Antonio: *Hacia una nueva estrategia en las relaciones Unión Europea América Latina.*
- **PP 04/05** Manalo, Rosario; Reyes, Melanie: *The MDGs: Boon or bane for gender equality and women's rights?*
- **PP 03/05** Fernández, Rafael: *Irlanda y Finlandia: dos modelos de especialización en tecnologías avan-zadas.*
- **PP 02/05** Alonso, José Antonio; Garcimartín, Carlos: *Apertura comercial y estrategia de desarrollo.*
- **PP 01/05** Lorente, Maite: *Diálogos entre culturas: una reflexión sobre feminismo, género, desarrollo y mujeres indígenas kichwuas.*
- **PP 02/04** Álvarez, Isabel: *La política europea de I+D: Situación actual y perspectivas.*
- **PP 01/04** Alonso, José Antonio; Lozano, Liliana; Prialé, María Ángela: *La cooperación cultural española: Más allá de la promoción exterior.*